ORIGINAL RESEARCH ARTICLE



Cost-Effectiveness Analysis of Abrocitinib Compared with Other Systemic Treatments for Severe Atopic Dermatitis in Spain

Rosa María Romero Jiménez¹ · Pedro Herranz Pinto² · Minia Campos Domínguez¹ · Susana Aceituno Mata³ · Alba Bellmunt³ · Miriam Prades³ · Daniel Arumi⁴ · Irene Hernández-Martín⁴ · Valeria Herrera-Lasso⁴ · Noelia Llevat⁴ · Alfonso De Lossada Juste⁴ · Francisco José Rebollo Laserna⁴

Accepted: 22 November 2023 © The Author(s) 2024

Abstract

Introduction Atopic dermatitis (AD) is a chronic, inflammatory skin disease characterized by itchy, painful, and dry skin. Despite the great number of available therapies, economic evaluations are still needed to provide evidence on their cost efficiency. This research aimed to evaluate the cost effectiveness of the Janus kinase (JAK) inhibitor abrocitinib (200 mg) compared with dupilumab (300 mg), tralokinumab (300 mg), baricitinib (2 and 4 mg), and upadacitinib (15 and 30 mg) for the treatment of patients with severe AD from the Spanish National Health System (NHS) perspective.

Methods A hybrid model consisting of a decision tree linked to a Markov model was developed to estimate costs, qualityadjusted life-years (QALYs), total years in response and incremental cost-per-QALY gained (willingness-to-pay [WTP] threshold: $\in 25,000/QALY$). Adults with severe AD entered the decision tree and response (75% reduction in baseline Eczema Area and Severity Index score, EASI-75) was considered at 16 and 52 weeks. After this time, patients entered the Markov model (remainder of the 10-year time horizon), which consisted of three health states: maintenance with active therapy, subsequent treatment, or death. All costs were presented in 2022 euros (\in). Additionally, cost per number-needed-to-treat (NNT) was calculated for abrocitinib and dupilumab based on a head-to-head post-hoc analysis.

Results Abrocitinib 200 mg was dominant (i.e., lower incremental costs and higher incremental benefit) compared with all studied alternatives (dupilumab 300 mg, tralokinumab 300 mg, baricitinib 2 and 4 mg, upadacitinib 15 and 30 mg) with a QALYs gain of 0.49, 0.60, 0.64, 0.43, 0.45, and 0.08, respectively, and per-person costs savings of \notin 22,097, \notin 24,140, \notin 14,825, \notin 7,116, \notin 12,805, and \notin 45,189, respectively. Considering the WTP threshold, abrocitinib was dominant or cost effective compared with all alternatives for most simulations. Additionally, abrocitinib was dominant compared with all alternatives when evaluating the cost effectiveness over a 5-year time horizon. NNT showed that abrocitinib was dominant versus dupilumab. **Conclusions** The results of the study show that abrocitinib is a cost-effective therapy compared with other JAK inhibitors and biological therapies from the Spanish NHS perspective.

Francisco José Rebollo Laserna francisco.rebollo@pfizer.com

- ¹ Hospital General Universitario Gregorio Marañón, Madrid, Spain
- ² Hospital Universitario La Paz, Madrid, Spain
- ³ Outcomes'10 S.L., Castellón de la Plana, Castellón, Spain
- ⁴ Pfizer SLU, Av. de Europa, 20, B, 28028 Alcobendas, Madrid, Spain

Key Points for Decision Makers

The analysis reveals that, when compared with other JAK inhibitors (baricitinib and upadacitinib) and biological treatment alternatives (dupilumab and tralokinumab), abrocitinib is a dominant therapy for the treatment of severe atopic dermatitis (AD) for the Spanish NHS, with lower costs and higher clinical benefits.

Abrocitinib's cost effectiveness, alongside its high efficacy and favorable safety profile, demonstrate its value as a treatment for severe AD.

1 Introduction

Atopic dermatitis (AD) is a common chronic, relapsing inflammatory skin disease characterized by itchy, painful, and dry skin that affects up to 15% of children and 5% of adults worldwide [1, 2]. The severe form of the disease can affect up to 10% of that population [2]. In Spain, the estimated prevalence of severe AD in the general population aged \geq 6 years is 0.10% [3]. Clinical manifestations of AD (mainly pruritus and pain) can negatively impact patients' quality of life, leading to impaired sleep and subsequent daytime tiredness and irritability, especially in those with severe forms of the disease [4, 5]. In fact, patients with AD frequently report negative effects on physical, emotional, and social life domains [6, 7]. In addition, AD has been associated with other atopic diseases, including asthma, allergic rhinitis, and food allergies [7]. Overall, AD imposes a high burden that might increase associated economic costs and healthcare resource utilization [4, 5]. No laboratory test for the diagnosis of AD exists. Instead, AD is diagnosed by clinical examination and its severity is classified with validated clinical tools like the Eczema Area and Severity Index (EASI) [8].

AD management aims to reduce symptoms and severity and improve long-term disease control [6]. Classical therapies for AD include emollients, topical corticosteroids and calcineurin inhibitors, and phototherapy, which have been the main treatment options for decades [9]. Moderate-tosevere cases require systemic therapies with immunosuppressants or corticosteroids [10]; however, their use is not indicated in the long term due to the emergence of adverse effects [10, 11]. The growing knowledge of the disease at a molecular level has led to the development of biological therapies targeting type 2 immune pathways IL-4 and IL-13 cytokines [7]. In this context, the monoclonal antibody dupilumab, the first biological therapy approved for treating patients with AD, has shown to have a positive benefit-risk profile [12]. However, responses to the therapy in some patients are not satisfactory [13], or patients may develop adverse events and need to discontinue. Other biologics targeting only IL-13, such as tralokinumab, have also been developed and are now approved in Europe [14]. As an alternative to biological therapies, the Janus kinase (JAK) inhibitors abrocitinib [15, 16], baricitinib [17], and upadacitinib [18–20] have been recently approved for use in moderate-to-severe AD, offering some advantages over biological therapies as they are orally available and have a rapid onset of action, especially with itch relief, and have potential to reach superior efficacy on endpoints compared with dupilumab [9].

Given the myriad of available therapies for severe AD, economic evaluations are needed to provide evidence on

efficiency and help the National Health System (NHS) in decision making. The aim of this research was to evaluate, from the perspective of the Spanish NHS, the cost effectiveness of the JAK inhibitor abrocitinib (200 mg) compared with dupilumab (300 mg), tralokinumab (300 mg), baricitinib (2 and 4 mg), and upadacitinib (15 and 30 mg) for the treatment of patients with severe AD who showed an inadequate response or inability to tolerate topical treatments or required systemic treatment to control the disease. Additionally, cost per number-needed-to-treat (NNT) for abrocitinib and dupilumab were calculated based on a head-to-head study, as this is a widely used tool in medical decision making and may increase understanding and relevance of costeffectiveness analysis findings [21].

2 Methods

2.1 Model Structure

The cost-effectiveness analysis was conducted using a hybrid model composed of a decision tree that captures short-term outcomes followed by a Markov model programmed in Excel, with a 10-year time horizon, to capture long-term outcomes. The structure of the model aims to capture the long-term treatment of a chronic condition such as AD. The design and implementation of the model was significantly informed by published literature [22] and other publicly available documentation reviewing and critiquing recent pharmacoeconomic modeling of treatments for AD [23, 24].

The model evaluated the costs and outcomes of adults with severe AD inadequately controlled with topical therapy or for whom topical therapies were medically inadvisable or required systemic treatment to control the disease. Patients entered the model through the decision tree (Fig. 1a), where they received abrocitinib or one of the comparators in combination with topical drug therapy. Response to treatment was assessed at 16 weeks as most of the included trials assessed their primary efficacy endpoint in this time period. Treatment response was defined by $a \ge 75\%$ reduction in EASI score (EASI-75) after a patient began treatment. If patients responded to treatment and did not discontinue, they continued receiving the same intervention until week 52, when the response was again evaluated. If patients did not respond to treatment at any node of the decision tree, they were considered to stop treatment and start subsequent therapy.

After 52 weeks, patients entered the Markov model (Fig. 1b) for long-term maintenance treatment, which consisted of three health states: maintenance with active therapy (abrocitinib or a comparator), subsequent treatment (in case of discontinuation or loss of response), or death. The simulation was conducted on a 6-month cycle length (with



half-cycle correction) for the remainder of the time horizon. Patients who were still administered abrocitinib or a comparator following the first 52 weeks of treatment were treated continuously with that product until loss of response or treatment discontinuation (response was re-assessed after each cycle). When this occurred, patients transitioned to subsequent treatment, where they remained until death. Transition to death state (absorbing state) was possible from any health state.

The modelled population had a mean age of 34 years at the start of the model, based on the mean age of participants in abrocitinib studies [25]. The model considered age-dependent mortality data for the Spanish population obtained from the National Institute of Statistics mortality tables [27]. The presence of AD did not increase the likelihood of death.

All data inputs were validated by a panel including three Spanish clinical experts (two dermatologists and one hospital pharmacist) through an advisory board and follow-up one-to-one consultations [28].

2.2 Efficacy

Table 1 Clinical inputs

Key efficacy inputs used in the model included the time to onset of response, the response at 16 and 52 weeks (decision tree), the annual loss of treatment response (Markov model), and the discontinuation both in the year of treatment initiation and each year after that (decision tree and Markov model).

As described above, the response was defined as EASI-75 score (Table 1). The model assumed patients in treatment

with JAK inhibitors begin experiencing therapeutic benefits at 8 weeks, as supported by clinical trial results that suggest response to abrocitinib commonly emerges within 4 weeks and has typically plateaued by 8 weeks [29]. For injectable medications (dupilumab and tralokinumab), it was assumed responders begin experiencing benefits at 16 weeks, as supported by clinical studies [29, 30].

Response rates are detailed in Table 1. The percentage of responders at 16 weeks was derived from a network meta-analysis for abrocitinib 200 mg, dupilumab 300 mg, tralokinumab 300 mg, and baricitinib 2 and 4 mg [31]. For upadacitinib 15 and 30 mg, response rates were derived from the AD-Up study [19].

Loss of treatment response beyond 52 weeks was assumed to occur at the same rate observed between 16–52 weeks (derived with reference to the proportion of week 16

Treatment	Measure	References
Response rate at 16 weeks ^a		
Abrocitinib 200 mg	74.3%	Assumptions based on Silverberg
Dupilumab 300 mg	61.5%	et al. [31] and JADE COMPARE
Tralokinumab 300 mg	49.3%	(data on file) [26]
Baricitinib 2 mg	41.3%	
Baricitinib 4 mg	47.3%	
Upadacitinib 15 mg	67.7%	Assumptions based on Reich et al.
Upadacitinib 30 mg	80.7%	[19] and JADE COMPARE (data on file) [26]
Sustained response at 52 weeks among 1	16-week responders	
Abrocitinib 200 mg	94.7%	JADE EXTEND (data on file) [32]
Dupilumab 300 mg	82.1%	NICE Dupilumab [35]
Tralokinumab 300 mg	82.1% ^b	Assumption
Baricitinib 2 mg	82.9% ^c	Assumption
Baricitinib 4 mg	92.1% ^d	Assumption
Upadacitinib 15 mg	78.6%	Silverberg et al. [36]
Upadacitinib 30 mg	89.5%	Silverberg et al. [36]
Annual discontinuation probability		
16–52 weeks	6.9%	JADE EXTEND (data on file) [32]
52 weeks+	6.3%	CADTH Dupixent [23]
Utility weights		
Baseline utility	0.6156	Simpson et al. [34]
Treatment responders	0.8772	Simpson et al. [34]
Treatment non-responders	0.7777	Simpson et al. [34]
Subsequent treatment	0.6084	Simpson et al. [34]

^aEfficacy data for the subgroup of adult patients with severe AD were only available for abrocitinib and dupilumab [26]. For comparators for which disaggregated data were unavailable for the adult-severe AD profile, the observed proportion of adult patients and change in response observed between patients with moderate-severe AD and those with severe AD was applied to the response rates obtained from the network meta-analysis (NMA) [31] or Reich et al. [19], based on JADE COMPARE [26]. For baricitinib and upadacitinib, the change observed for abrocitinib was applied, and for tralokinumab, the change from dupilumab was applied.

^bThe same rate as for dupilumab is assumed

^cAssumed to be the mean between the rate of abrocitinib 100 mg and upadacitinib 15 mg

^dAssumed to be the mean between the rate of abrocitinib 200 mg and upadacitinib 30 mg

responders who sustained response at week 52 based on the clinical trials of each comparator).

2.3 Treatment Discontinuation

Due to the lack of treatment discontinuation rates for each treatment, it was assumed to be the same for all comparators: 6.9% of patients discontinued during the first 52 weeks (based on rates observed for EASI-75 responders discontinuing abrocitinib 200 mg in JADE EXTEND) [32], and 6.3% discontinued each subsequent year (based on results from LIBERTY AD SOLO for dupilumab) [23].

2.4 Utility Values

The analysis applied health state utilities based on EuroQol instrument (EQ-5D-3L) data collected in two dupilumab phase III clinical trials for adults with moderate-to-severe AD [33] (Table 1), in line with NICE recommendations [24]. EQ-5D-3L is a standardized health-related quality of life (HRQoL) questionnaire which evaluates five dimensions (mobility, self-care, usual activities, pain/discomfort, and anxiety/depression) in three levels of severity. Utilities reported range from 0 (death) to 1 (perfect health).

In addition, the model applied adjustment factors that accounted for the HRQoL impact of Spanish cohort aging [34]. Population norms were applied multiplicatively to treatment state utilities to reflect the aging of the cohort beyond its average age upon initiation of treatment and the resultant decline in baseline utility over time.

2.5 Cost

Costs included drug-related costs (acquisition and administration), adverse events (AEs) management, testing, medical visits and hospitalizations, and cost of subsequent treatment. All costs were presented in 2022 euros (\in).

Drug costs were calculated using list prices obtained from the database of the Official College of Pharmacists [37] (the rebate established in Spanish Royal Decree-Law 8/2010 was applied [37]). In addition, a one-time cost of training patients to self-administer injectable products (specialized nursing visit) was included for dupilumab and tralokinumab (Table 2). No administration costs were considered for oral therapies.

Annual testing was assumed to be the same for all comparators. For AE management, the model included those experienced by at least 5% of participants in clinical trials for any comparator [29, 38–41]. Frequencies of AEs were annualized (Table S1, see electronic supplementary material [ESM]). AEs were assumed to be usually managed through a visit to a dermatologist or an ophthalmologist for allergic conjunctivitis (Table 2). The model also accounted for the costs of primary care visits, emergency room visits, and hospitalizations depending on treatment response. The average annual utilization of each resource was derived from a longitudinal, non-interventional, retrospective cohort study in Canadian patients with AD [42], and Spanish unitary costs [43] were applied (Table 2).

For subsequent treatment, experts assumed an average cost between biological drugs and JAK inhibitors, considering the highest price of each category, in order not to underestimate the cost of this state (Table 2).

2.6 Model Outputs

The model reported outputs including total costs, years in response, years in subsequent therapy and quality-adjusted life-years (QALYs). The incremental cost per QALY gained was calculated. A willingness-to-pay (WTP) threshold of \notin 25,000 per QALY gained was considered in line with the Spanish guidelines [44]. If the incremental cost-effectiveness ratio (ICER) of the treatment was below the WTP threshold, then it was considered a cost-effective alternative. When the intervention was both clinically superior and cost saving, it was referred to as an economically 'dominant' strategy. The opposite was a 'dominated' strategy. Another option is that an alternative was less costly, but also less effective.

Both costs and outcomes were discounted at 3% per year according to the local recommendations for economic evaluation of health technologies [45, 46].

2.7 Sensitivity Analyses

One-way sensitivity analysis (OWSA), scenario and probabilistic sensitivity analyses (PSA) were carried out to identify the most influential parameters and test the robustness of the results.

OWSA was performed to investigate the impact of individual model parameters used in the base-case analysis on model outcomes, using the hypothetical increases or decreases of 20%. Finally, results were compared with the base case in a tornado diagram.

The PSA was conducted using a Montecarlo simulation with 1000 iterations. A PSA simultaneously sets all inputs to a value randomly sampled from the appropriate distribution (Table S2, see ESM). When uncertainty data were not reported, the standard error was assumed to be 10% of the mean.

A scenario analysis was performed for a 5-year time horizon. To further account for the structural uncertainty, alternative scenarios were constructed to assess the impact of variations in discontinuation rates on cost effectiveness (i.e. maintaining the values of abrocitinib for 16–52 weeks and dupilumab for +52 weeks and increasing the other values

Cost type	Unit cost	Moi	thly cost	Loading	dose cost	Administration cost	References
Drug costs		1					
Abrocitinib 200 mg	€31.73	€96	5.71				BotPlus
Dupilumab 300 mg	€560.34	€12	18.24	€1120.6	8	€72.06	[37],
Tralokinumab 150 mg	€280.17	€12	18.24	€1120.6	8	€72.06	eSalud
Baricitinib 2 mg	€25.77	€78-	4.37				[45]
Baricitinib 4 mg	€25.77	€78-	4.37				
Upadacitinib 15 mg	€31.08	€94	5.00				
Upadacitinib 30 mg	€62.16	€18	92.00				
Subsequent treatment (monthly cost) ^a	€1555.12						Assumption
Test (annual cost) ^b	€394.28						eSalud [43]
AE management	€83.86 (ophtha AE	lmology	visit) for allerg	ic conjunctivit	is/€74.68 (derma	tology visit) for all other	eSalud [43]
Visits and hospitalizations by type of response	Responders util (annual)	ization	Non-respond (annual)	ler utilization	Cost per visit		
Hospitalization	0.30		0.50		€3059.93	Data on file [42], eSalud [43]
Emergency room visit	0.60		1.30		€268.38		
Primary care visits	9.80		21.20		€41.60		

AE adverse event

Table 2 Cost inputs

^aAverage cost of JAK inhibitor and biologic of highest price

^bAnnual testing includes: 1 renal function test, 1 lipid profile, 1 complete blood count, 1 liver function test, 1 tuberculosis test and 1 hepatitis B and C test. Same costs considered for all the pharmacological therapies included in the model

by 10% [scenario 1] or decreasing the other values by 2% [scenario 2]; using the same values for the same drug type, and increasing or decreasing the others by 2% [scenario 3 and 4]; Table S3, see ESM).

2.8 Number-Needed-to-Treat (NNT) and Cost Per Responder Analysis

The NNT is an outcome measure commonly used in clinical settings, providing a quick, short-hand approach to estimating relative efficacy of different treatments [47]. The NNT

of the cost per NNT for abrocitinib 200 mg and dupilumab 300 mg based on EASI-75 responders from a post-hoc analysis of patients with severe AD from the JADE COMPARE study [21, 49], as these are the unique treatments that have been compared in a head-to-head study. Firstly, the NNT for achieving an EASI-75 response was obtained using the difference in response for active treatment (abrocitinib 200 mg or dupilumab 300 mg) versus placebo at 12 and 16 weeks, where the NNT is calculated as the inverse of the probability of response to the active treatment minus the probability of response to placebo.

 $NNT = \frac{1}{(probability of treatment response - probability of placebo response)}$

corresponds to the average number of patients who need to be treated with a particular therapy to achieve one extra positive outcome when compared with another therapy or placebo. The ideal NNT is 1 because it implies every patient treated will achieve the stated clinical benefit. The higher the NNT, the less effective the intervention [48].

NNT is often used as a tool in medical decision-making [47]. For this reason, we additionally included an analysis

The cost per NNT for each drug was obtained by multiplying the annual cost of each therapy over the first year of treatment by its NNT (at 12 and 16 weeks).

Additionally, NNT for abrocitinib versus dupilumab was calculated. An incremental cost per additional patient with clinically meaningful outcome was calculated as the difference in annual drug costs (abrocitinib vs dupilumab) multiplied by the corresponding NNT. When the difference in total cost is negative and the NNT is positive, the incremental cost per additional patient is denoted as 'dominant'.

3 Results

3.1 Base-Case Results

In the base-case analysis, abrocitinib 200 mg was dominant versus all alternatives (dupilumab 300 mg, tralokinumab 300 mg, baricitinib 2 mg, baricitinib 4 mg, upadacitinib 15 mg, and upadacitinib 30 mg), generating a QALYs gain of 0.49, 0.60, 0.64, 0.43, 0.45, and 0.08 with cost savings of ϵ 22,097.46, ϵ 24,139.97, ϵ 14,825.26, ϵ 7116.25, ϵ 12,805.37, and ϵ 45,189.39, respectively. Moreover, abrocitinib 200 mg led to an increase in response time versus comparators of 2.08, 2.55, 2.75, 1.84, 2.00, and 0.38 years, respectively. Table 3 contains the detailed base-case results.

3.2 Sensitivity Analyses

Sensitivity analyses confirmed the robustness of the results. Changes in parameters analyzed in the OWSA did not affect the results of dominance. The PSA showed that abrocitinib was dominant or cost effective versus all comparators for most of the simulations performed, considering the WTP threshold of €25,000 per QALY gained (Table 4).

3.3 Scenario Analyses

When evaluating the cost effectiveness over a 5-year time horizon, abrocitinib was dominant versus all comparators (dupilumab 300 mg, tralokinumab 300 mg, baricitinib 2 mg, baricitinib 4 mg, upadacitinib 15 mg, and upadacitinib 30 mg), generating a QALYs gain of 0.27, 0.35, 0.37, 0.27, 0.22, and 0.01 with per-person cost savings of \in 13,984, \in 15,503, \in 7910, \in 3867, \in 6023, and \in 30,839, respectively. Table 5 describes the model results for a 5-year time

Table 3 Base-case results

Results	Abrocitinib 200 mg	Dupilumab 300 mg	Tralokinumab 300 mg	Baricitinib 2 mg	Baricitinib 4 mg	Upadacitinib 15 mg	Upadacitinib 30 mg
Costs							
Pharmacological	€132,241	€152,652	€154,389	€144,947	€137,906	€143,507	€177,023
Drug administration	€0	€72	€72	€0	€0	€0	€0
Adverse events	€518	€487	€423	€492	€520	€526	€657
Testing costs	€1625	€968	€799	€702	€1002	€965	€1507
Medical visits and hospitalization	€18,883	€21,186	€21,724	€21,951	€20,956	€21,074	€19,270
Total costs	€153,267	€175,365	€177,407	€168,092	€160,383	€166,073	€198,457
Total QALYs	6.33	5.84	5.73	5.70	5.90	5.88	6.25
Years in response	4.43	2.35	1.89	1.68	2.59	2.43	4.05
Years in subsequent therapy	5.55	7.62	8.09	8.30	7.39	7.55	5.93
$\Delta Cost$ (abrocitinib vs)		- €22,097	- €24,140	- €14,825	- €7116	- €12,805	- €45,189
Δ QALYs (abrocitinib vs)		0.49	0.60	0.64	0.43	0.45	0.08
ICER (abrocitinib vs)		Dominant	Dominant	Dominant	Dominant	Dominant	Dominant

ICER incremental cost-effectiveness ratio, QALYs quality-adjusted life-years

Table 4PSA results of the base case

Results	Abrocitinib 200 mg vs						
	Dupilumab 300 mg	Tralokinumab 300 mg	Baricitinib 2 mg	Baricitinib 4 mg	Upadacitinib 15 mg	Upadacitinib 30 mg	
Dominant	88.2%	88.8%	85.5%	70.2%	86.4%	60.9%	
Cost effective	0.4%	0.7%	3.4%	11.7%	1.7%	0.0%	
Not cost effective	0.0%	0.0%	0.3%	5.9%	0.7%	0.0%	
Less costly, less effective	11.4%	10.4%	10.5%	8.6%	10.3%	39.1%	
Dominated	0.0%	0.1%	0.3%	3.6%	0.9%	0.0%	

PSA probabilistic sensitivity analysis

horizon. PSA results demonstrated the robustness of the analysis (Table S4, see ESM).

Table 6 presents the results of the scenarios related to changes in discontinuation values. In all scenarios, abrocitinib was found to be dominant over all alternatives.

3.4 NNT Analysis

The cost per NNT of abrocitinib and dupilumab versus placebo is shown in Table 7. Abrocitinib had the lowest NNT compared with placebo to achieve an additional EASI-75 responder for both week 12 and week 16, demonstrating its sustainability of the response. As the EASI-75 response time increases, costs per NNT increase for both therapies, being lower for abrocitinib than for dupilumab in both cases.

When compared with dupilumab, the annual cost of abrocitinib was lower ($-\epsilon4,151.04$), with an NNT for EASI-75 responders of 4.37 for week 12 responders and 8.00 for

week 16 responders, so abrocitinib was dominant compared with dupilumab as it was associated with lower costs and favorable clinical outcomes (Table 8).

4 Discussion

This analysis has shown that abrocitinib 200 mg is dominant (i.e., lower incremental costs and higher incremental benefit) compared with currently available alternatives, with QALY gains ranging from 0.08 to 0.64 and costs savings of ϵ 7116.25– ϵ 45,189.39. The cost difference was mainly explained by lower abrocitinib acquisition costs and lower medical visits and hospitalization costs for patients treated with abrocitinib. Regarding differences in clinical results, abrocitinib significantly increased time in response versus comparators, delaying the loss of response and the transition to subsequent therapy, and increasing QALYs.

Table 5 5-year time horizon deterministic results

Results	Abrocitinib 200 mg	Dupilumab 300 mg	Tralokinumab 150 mg	Baricitinib 2 mg	Baricitinib 4 mg	Upadacitinib 15 mg	Upadacitinib 30 mg
Total QALYs	3.54	3.27	3.18	3.16	3.27	3.31	3.52
Total costs	€78,280	€92,264	€93,783	€86,190	€82,147	€84,303	€109,120
ΔCost		- €13,984	- €15,503	- €7910	- €3867	- €6023	- €30,839
ΔQALYs		0.27	0.35	0.37	0.27	0.22	0.01
ICER		Dominant	Dominant	Dominant	Dominant	Dominant	Dominant

ICER incremental cost-effectiveness ratio, QALYs quality-adjusted life-years

Table 6Deterministic resultsof scenarios with differentdiscontinuation rates

Discontinua- tion scenario ^a Abrocitin Dupiluma 300 mg	Abrocitinib 200 mg vs							
	Dupilumab 300 mg	Tralokinumab 300 mg	Baricitinib 2 mg	Baricitinib 4 mg	Upadacitinib 15 mg	Upadacitinib 30 mg		
Scenario 1	Dominant	Dominant	Dominant	Dominant	Dominant	Dominant		
Scenario 2	Dominant	Dominant	Dominant	Dominant	Dominant	Dominant		
Scenario 3	Dominant	Dominant	Dominant	Dominant	Dominant	Dominant		
Scenario 4	Dominant	Dominant	Dominant	Dominant	Dominant	Dominant		

^aDiscontinuation scenarios: (1) maintaining the values of abrocitinib for 16–52 weeks (6.9%) and dupilumab for + 52 weeks (6.3%) and increasing the other values by 10%; (2) maintaining the values of abrocitinib for 16–52 weeks (6.9%) and dupilumab for + 52 weeks (6.3%) and decreasing the other values by 2%; (3) using the same values for the same drug type, and increasing the others by 2%; (4) using the same drug type, and decreasing the others by 2%

Table 7	NNT and cost per
NNT by	treatment response
assessm	ent period

Drug	Week 12 responders		Week 16 responders		
	NNT EASI-75 (95% CI)	Cost per NNT	NNT EASI-75 (95% CI)	Cost per NNT	
Abrocitinib 200 mg	1.51 (1.28–1.73)	€17,478.91	1.88 (1.51–2.32)	€21,742.06	
Dupilumab 300 mg	2.30 (1.79–3.05)	€36,266.27	2.45 (1.85-3.38)	€38,577.35	

CI confidence interval, EASI-75 75% reduction in baseline Eczema Area and Severity Index score, NNT number needed to treat

Drug	NNT EASI-75 (95% CI) week 12 responders	NNT EASI-75 (95% CI) week 16 responders	Cost difference	Incremental cost per additional patient with outcome
Abrocitinib vs dupilumab	4.37 (2.78–9.19)	8.00 (3.94–587.81)	- €4151.04	Dominant

Table 8 NNT and incremental costs per patient comparing abrocitinib with dupilumab

CI confidence interval, EASI-75 75% reduction in baseline Eczema Area and Severity Index score, NNT number needed to treat

The probabilistic analysis confirmed deterministic results while allowing for full, simultaneous parameter variation. In addition, the scenario analyses confirmed the base-case results, demonstrating that abrocitinib is a dominant alternative in Spain, even with shorter time horizons or different discontinuation rates.

To the best of our knowledge, this is the first study to evaluate the cost effectiveness of abrocitinib versus currently available therapies for treating AD in Spain. This is also the first study to date to compare the cost per NNT of a JAK inhibitor versus a biologic for severe AD in Spain. As we have seen, treating severe AD patients with abrocitinib would result in more patients with improved outcomes at a lower cost. This calculation of cost per unit of clinical effectiveness is an approach clinicians and healthcare decision makers understand well. However, the value of this measure cannot be placed in relation to an established threshold of how much it is reasonable to invest per health outcome, unlike in cost-effectiveness models. It is therefore useful to have both methodologies available to complement health decisions [50].

Abrocitinib economic evaluation data are still scarce in the literature. In a previous study conducted in the US, the cost effectiveness of abrocitinib (using an average of the net prices of baricitinib and upadacitinib as a placeholder) versus the standard of care and dupilumab in patients with moderate-to-severe AD was evaluated; the authors reported a cost per QALY gained for the base case of US\$148,300 for abrocitinib compared with the standard of care, resulting in a cost-effective alternative at a threshold of US\$100,000-US\$150,000. However, abrocitinib was not cost effective (US\$303,400 per QALY gained) compared with dupilumab [51, 52]. It is worth noting that marketed prices for abrocitinib were not available at the time of the study. Moreover, cost effectiveness was estimated using a Markov model previously developed for dupilumab [51], which could explain, at least in part, the differences observed with respect to our study. In contrast, the analysis conducted by the Canadian Agency for Drugs and Technologies in Health showed that abrocitinib 200 mg was dominant versus dupilumab for the treatment of adults with moderateto-severe AD, in line with our results [53]. Another study reported a post-hoc economic analysis of abrocitinib versus placebo or dupilumab using data from the clinical trials JADE MONO-2 [16] and JADE COMPARE [29], showing a reduction in overall work impairment and associated costs at week 12 in patients treated with abrocitinib compared with those receiving placebo, as reported in JADE MONO-2. Moreover, the mean number of physician visits in JADE COMPARE at week 16 was lower for abrocitinib (0.9) versus dupilumab (1.3) and placebo (1.6) [54]. These results are in line with the lower medical visits and hospitalization costs of abrocitinib versus dupilumab in our model.

A recent meta-analysis showed that abrocitinib is one of the most effective alternatives in improving Investigator's Global Assessment (IGA) and EASI scores while being a safe option for treating patients with moderate-to-severe AD [55, 56]. In line with this evidence, our analysis showed that patients treated with abrocitinib were more years in response than those treated with the alternatives. Regarding the impact of AD on patients' lives, recently published studies analyzed patient-reported outcomes in adults and adolescents with moderate-to-severe AD and showed that abrocitinib improved AD symptoms (itch severity, pain, erythema), sleep disturbance, depression, and anxiety, increasing general health-related quality of life in these patients [57, 58]. Consistent with these studies, our analysis showed that patients who received abrocitinib obtained a better response to treatment, translating into a higher number of QALYs and, therefore, a better quality of life.

Our model is subject to limitations. Firstly, efficacy data for the severe subgroup was not available for all treatments, so it was estimated by applying ratios of overall versus subpopulation response based on original data from JADE COMPARE of abrocitinib for JAK inhibitor and from dupilumab for tralokinumab. Moreover, response rates at 16 weeks for upadacitinib were not available from the network meta-analysis (NMA), so clinical trial data were used. However, the response rates of each drug were varied in the OWSA and had no significant impact on abrocitinib dominance. Secondly, treatment discontinuation was assumed to be the same for all treatments. However, a sensitivity analysis was performed including variations in discontinuation rates, noting that it does not impact dominance results. Thirdly, the model does not explicitly account for treatment sequencing; it assumes use of a basket of treatments (subsequent treatments), the cost of which is calculated as the mean between the JAK inhibitor and the biologic with the highest price. Nevertheless, the model aims not to compare sequences of treatment but to compare abrocitinib against its comparators in the same treatment line. Furthermore, applying the same cost of subsequent therapy for all treatments, the only thing that influences it is the length of time patients are on subsequent therapy, which is lower for abrocitinib (as patients are on response longer). Another limitation is that resources consumption used for the estimation of medical visits and hospitalizations were derived from a Canadian study. Nevertheless, the expert panel validated the annual number of visits as representative of the Spanish environment. Finally, health state utilities were derived from two clinical trials for adults with AD but were not Spanish data. However, these health-state utilities are in line with NICE recommendations and were validated by Spanish clinical experts.

5 Conclusion

Abrocitinib is a dominant therapy compared with other JAK inhibitors and biological therapies for the Spanish NHS. These results might help decision making at a regulatory level regarding the use of abrocitinib in routine clinical practice.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s41669-023-00459-2.

Acknowledgements The authors would like to thank Carmen Peral, Pfizer employee, for her collaboration in the initial development of this research.

Declarations

Funding This study was sponsored by Pfizer, S.L.U., Spain.

Conflicts of interest This analysis was sponsored by Pfizer, S.L.U., Spain. Arumi, Hernández-Martín, Herrera-Lasso, Llevat, De Lossada Juste, and Rebollo Laserna are employees of Pfizer (Spain). Aceituno Mata, Bellmunt, and Prades disclose that they are paid consultants to Outcomes10, a company paid by Pfizer to assist with the writing of this manuscript. Herranz Pinto has served as investigator and/or speaker for Abbvie, Almirall, Bristol Myers Squibb, Janssen, Leopharma, Lilly, Novartis, Pfizer, Regeneron, Sandoz, Sanofi, and UCB pharma. Campos Dominguez has served as speaker for Sanofi and Abbvie and as investigator for Abbvie. Romero Jimenez declares no conflicts of interest.

Availability of data and material Data are available from the corresponding author upon reasonable request.

Ethics approval Not applicable.

Consent to participate Not applicable.

Consent for publication Not applicable.

Code availability The datasets generated and analyzed during the current study are available from the corresponding author on reasonable request.

Author contributions RMRJ, PHP, MCD validated the parameters and provided data about local management of patients. SAM, AB and MP performed the analysis. All authors participated in the interpretation of the results. The first draft of the manuscript was primarily written by SAM and AB, and all authors provided comment and revision on subsequent versions of the manuscript. All authors reviewed and approved the final manuscript.

Open Access This article is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License, which permits any non-commercial use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by-nc/4.0/.

References

- Silverberg JI, Feldman SR, Smith Begolka W, Johnson MB, Stanojev C, DiBonaventura M, et al. Patient perspectives of atopic dermatitis: comparative analysis of terminology in social media and scientific literature, identified by a systematic literature review. J Eur Acad Dermatol Venereol. 2022;36(11):1980–90.
- Weidinger S, Nosbaum A, Simpson E, Guttman E. Good practice intervention for clinical assessment and diagnosis of atopic dermatitis: findings from the atopic dermatitis quality of care initiative. Dermatol Ther. 2022;35(3): e15259.
- Sicras-Mainar A, Navarro-Artieda R, Armario-Hita JC. Severe atopic dermatitis in Spain: a real-life observational study. Ther Clin Risk Manag. 2019;15:1393–401.
- Holm JG, Agner T, Clausen ML, Thomsen SF. Quality of life and disease severity in patients with atopic dermatitis. J Eur Acad Dermatol Venereol. 2016;30(10):1760–7.
- Sicras-Mainar A, Navarro-Artieda R, Carrascosa Carrillo JM. Economic impact of atopic dermatitis in adults: a populationbased study (IDEA study). Actas Dermosifiliogr (Engl Ed). 2018;109(1):35–46.
- Weidinger S, Novak N. Atopic dermatitis. Lancet. 2016;387(10023):1109-22.
- Langan SM, Irvine AD, Weidinger S. Atopic dermatitis. Lancet. 2020;396(10247):345–60.
- Frazier W, Bhardwaj N. Atopic dermatitis: diagnosis and treatment. Am Fam Physician. 2020;101(10):590–8.
- 9. Paolino A, Alexander H, Broderick C, Flohr C. Non-biologic systemic treatments for atopic dermatitis: current state of the art and future directions. Clin Exp Allergy. 2023;53(5):495–510.
- Ferrucci SM, Tavecchio S, Marzano AV, Buffon S. Emerging systemic treatments for atopic dermatitis. Dermatol Ther (Heidelb). 2023;13(5):1071–81.
- Zimmermann M, Rind D, Chapman R, Kumar V, Kahn S, Carlson J. Economic evaluation of dupilumab for moderate-to-severe atopic dermatitis: a cost-utility analysis. J Drugs Dermatol. 2018;17(7):750–6.
- Simpson EL, Bieber T, Guttman-Yassky E, Beck LA, Blauvelt A, Cork MJ, et al. Two phase 3 trials of dupilumab versus placebo in atopic dermatitis. N Engl J Med. 2016;375(24):2335–48.

- Tampa M, Mitran CI, Mitran MI, Georgescu SR. A new horizon for atopic dermatitis treatments: JAK inhibitors. J Pers Med. 2023;13(3):384.
- Blauvelt A, Langley RG, Lacour JP, Toth D, Laquer V, Beissert S, et al. Long-term 2-year safety and efficacy of tralokinumab in adults with moderate-to-severe atopic dermatitis: interim analysis of the ECZTEND open-label extension trial. J Am Acad Dermatol. 2022;87(4):815–24.
- Simpson EL, Sinclair R, Forman S, Wollenberg A, Aschoff R, Cork M, et al. Efficacy and safety of abrocitinib in adults and adolescents with moderate-to-severe atopic dermatitis (JADE MONO-1): a multicentre, double-blind, randomised, placebocontrolled, phase 3 trial. Lancet. 2020;396(10246):255–66.
- 16. Silverberg JI, Simpson EL, Thyssen JP, Gooderham M, Chan G, Feeney C, et al. Efficacy and safety of abrocitinib in patients with moderate-to-severe atopic dermatitis: a randomized clinical trial. JAMA Dermatol. 2020;156(8):863–73.
- 17. Simpson EL, Lacour JP, Spelman L, Galimberti R, Eichenfield LF, Bissonnette R, et al. Baricitinib in patients with moderateto-severe atopic dermatitis and inadequate response to topical corticosteroids: results from two randomized monotherapy phase III trials. Br J Dermatol. 2020;183(2):242–55.
- Guttman-Yassky E, Teixeira HD, Simpson EL, Papp KA, Pangan AL, Blauvelt A, et al. Once-daily upadacitinib versus placebo in adolescents and adults with moderate-to-severe atopic dermatitis (Measure Up 1 and Measure Up 2): results from two replicate double-blind, randomised controlled phase 3 trials. Lancet. 2021;397(10290):2151–68.
- Reich K, Teixeira HD, de Bruin-Weller M, Bieber T, Soong W, Kabashima K, et al. Safety and efficacy of upadacitinib in combination with topical corticosteroids in adolescents and adults with moderate-to-severe atopic dermatitis (AD Up): results from a randomised, double-blind, placebo-controlled, phase 3 trial. Lancet. 2021;397(10290):2169–81.
- 20. Simpson EL, Papp KA, Blauvelt A, Chu CY, Hong HC, Katoh N, et al. Efficacy and safety of upadacitinib in patients with moderate to severe atopic dermatitis: analysis of follow-up data from the Measure Up 1 and Measure Up 2 randomized clinical trials. JAMA Dermatol. 2022;158(4):404–13.
- 21. Simpson EL, Silverberg JI, Thyssen JP, Viguier M, Thaci D, de Bruin-Weller M, et al. Efficacy and safety of abrocitinib in patients with severe and/or difficult-to-treat atopic dermatitis: a post hoc analysis of the randomized phase 3 JADE COMPARE trial. Am J Clin Dermatol. 2023;24(4):609–21.
- Kuznik A, Bégo-Le-Bagousse G, Eckert L, Gadkari A, Simpson E, Graham CN, et al. Economic evaluation of dupilumab for the treatment of moderate-to-severe atopic dermatitis in adults. Dermatol Therapy. 2017;7(4):493–505.
- Canadian Agency for Drugs and Technologies in Health (CADTH). SR0533 dupixent pharmacoeconomic review (CADTH). Available from: https://www.cadth.ca/sites/default/ files/cdr/pharmacoeconomic/SR0533_Dupixent_PE_Report. pdf. Accessed on July 2021.
- 24. NICE. Abrocitinib, tralokinumab or upadacitinib for treating moderate to severe atopic dermatitis. Available from: https:// www.nice.org.uk/guidance/ta814/resources/abrocitinib-tralo kinumab-or-upadacitinib-for-treating-moderate-to-severeatopic-dermatitis-pdf-82613310355141. Accessed in Oct 2023.
- 25. Cibinqo. Summary of product characteristics. Available from: https://www.ema.europa.eu/en/documents/product-information/ cibinqo-epar-product-information_en.pdf. Accessed on Oct 2023.
- 26. JADE COMPARE. Data on file. Pfizer.
- Instituto Nacional de Estadística (INE). Tablas de mortalidad por año, sexo, edad y funciones 2019. Available from: https://www. ine.es/jaxiT3/Tabla.htm?t=27153. Accessed on July 2021.

- 28. All data inputs were validated by an expert panel of three Spanish healthcare professionals (two dermatologists and one hospital pharmacist) through and advisory board in 2021 to ensure that the study perspective is consistent with current Spanish clinical practice.
- 29. Bieber T, Simpson EL, Silverberg JI, Thaçi D, Paul C, Pink AE, et al. Abrocitinib versus placebo or dupilumab for atopic dermatitis. N Engl J Med. 2021;384(12):1101–12.
- Wollenberg A, Blauvelt A, Guttman-Yassky E, Worm M, Lynde C, Lacour JP, et al. Tralokinumab for moderate-to-severe atopic dermatitis: results from two 52-week, randomized, double-blind, multicentre, placebo-controlled phase III trials (ECZTRA 1 and ECZTRA 2). Br J Dermatol. 2021;184(3):437–49.
- 31. Silverberg JI, Thyssen JP, Fahrbach K, Mickle K, Cappelleri JC, Romero W, et al. Comparative efficacy and safety of systemic therapies used in moderate-to-severe atopic dermatilis: a systematic literature review and network meta-analysis. J Eur Acad Dermatol Venereol.
- 32. JADE EXTEND ad-hoc análisis. Pfizer data on file. 2020.
- 33. Simpson EL. Dupilumab improves general health-related quality-of-life in patients with moderate-to-severe atopic dermatitis: pooled results from two randomized, controlled phase 3 clinical trials. Dermatol Ther (Heidelb). 2017;7(2):243–8.
- Janssen MF, Szende A, Cabases J, Ramos-Goni JM, Vilagut G, Konig HH. Population norms for the EQ-5D-3L: a cross-country analysis of population surveys for 20 countries. Eur J Health Econ. 2019;20(2):205–16.
- National Institute for Health and Care Excellence (NICE). [ID1048] Dupilumab for treating moderate to severe atopic dermatitis (Appraisal Consultation Document). Available from: https://www.nice.org.uk/guidance/ta534/documents/appraisalconsultation-document. Accessed on July 2021.
- Silverberg JI, de Bruin-Weller M, Bieber T, Soong W, Kabashima K, Costanzo A, et al. Upadacitinib plus topical corticosteroids in atopic dermatitis: week 52 AD Up study results. J Allergy Clin Immunol. 2022;149(3):977-87.e14.
- Consejo General de Colegios Oficiales de Farmacéuticos. Bot Plus Web. Available from: https://botplusweb.portalfarma.com/. Accessed on Jan 2023.
- Blauvelt A, de Bruin-Weller M, Gooderham M, Cather JC, Weisman J, Pariser D, et al. Long-term management of moderateto-severe atopic dermatitis with dupilumab and concomitant topical corticosteroids (LIBERTY AD CHRONOS): a 1-year, randomised, double-blinded, placebo-controlled, phase 3 trial. Lancet. 2017;389(10086):2287–303.
- Wollenberg A, Howell MD, Guttman-Yassky E, Silverberg JI, Kell C, Ranade K, et al. Treatment of atopic dermatitis with tralokinumab, an anti-IL-13 mAb. J Allergy Clin Immunol. 2019;143(1):135–41.
- 40. Reich K, Kabashima K, Peris K, Silverberg JI, Eichenfield LF, Bieber T, et al. Efficacy and safety of baricitinib combined with topical corticosteroids for treatment of moderate to severe atopic dermatitis: a randomized clinical trial. JAMA Dermatol. 2020;156(12):1333–43.
- Guttman-Yassky E, Thaçi D, Pangan AL, Hong HC, Papp KA, Reich K, et al. Upadacitinib in adults with moderate to severe atopic dermatitis: 16-week results from a randomized, placebocontrolled trial. J Allergy Clin Immunol. 2020;145(3):877–84.
- 42. Pfizer. Health care resource utilization and burden associated with atopic dermatitis; analyses with the RAMQ Database (Non-Interventional (NI) Final Study Report No. B7451065). 2020.
- 43. Gisbert R, Brosa M. Base de datos de costes sanitarios y ratios coste-efectividad españoles: eSalud. Barcelona: Oblikue Consulting, S.L. Available from: http://www.oblikue.com/bddcostes/. Accessed on Jan 2023.

- Sacristán JA, Oliva J, Campillo-Artero C, Puig-Junoy J, Pinto-Prades JL, Dilla T, et al. ¿Qué es una intervención sanitaria eficiente en España en 2020? Gac Sanit. 2020;34(2):189–93.
- 45. Grupo Genesis. Guía de evaluación económica e impacto presupuestario en los informes de evaluación de medicamentos. Available from: https://gruposdetrabajo.sefh.es/genesis/genesis/Docum ents/GUIA_EE_IP_GENESIS-SEFH_19_01_2017.pdf. Accessed on July 2021.
- 46. López-Bastida J, Oliva J, Antoñanzas F, García-Altés A, Gisbert R, Mar J, et al. Spanish recommendations on economic evaluation of health technologies. Eur J Health Econ. 2010;11(5):513–20.
- Garg V, Shen X, Cheng Y, Nawarskas JJ, Raisch DW. Use of number needed to treat in cost-effectiveness analyses. Ann Pharmacother. 2013;47(3):380–7.
- 48. Manriquez JJ, Villouta MF, Williams HC. Evidence-based dermatology: number needed to treat and its relation to other risk measures. J Am Acad Dermatol. 2007;56(4):664–71.
- 49. FC01.05 Silverberg J, Simpson E, Thyssen JP, Viguier M, Thaçi D, de Bruin-Weller M. Efficacy of abrocitinib in severe and difficult-to-treat patients with moderate-to-severe atopic dermatitis in the phase 3 JADE COMPARE study. In: Presented at the 30th EADV Virtual Meeting.
- 50. Nunez M, Huete T, de la Cueva P, Sacristan JA, Hartz S, Dilla T. A cost-per-number needed to treat analysis assessing the efficiency of biologic drugs in moderate to severe plaque psoriasis. Actas Dermosifiliogr (Engl Ed). 2019;110(7):546–53.
- Agboola F, Atlas SJ, Brouwer E, Carlson JJ, Hansen RN, Herron-Smith S, et al. JAK inhibitors and monoclonal antibodies for the treatment of atopic dermatitis: effectiveness and value. J Manag Care Spec Pharm. 2022;28(1):108–14.
- 52. Atlas SJ, Brouwer E, Fox G, Carlson J, Campbell JD, Agboola F, et al. Institute for clinical and economic review; JAK inhibitors

and monoclonal antibodies for the treatment of atopic dermatitis: effectiveness and value; evidence report. 2021. Available from: https://icer.org/assessment/atopic-dermatitis-2021/#timeline.

- 53. Canadian Agency for Drugs and Technologies. CADTH reimbursement recommendation (draft) Abrocitinib (Cibinqo). 2022.
- 54. Gooderham MJ, Chu CY, Rojo R, Valdez H, Biswas P, Cameron MC, et al. Economic impact of abrocitinib monotherapy and combination therapy in patients with moderate-to-severe atopic dermatitis: results from JADE MONO-2 and JADE COMPARE. JAAD Int. 2021;4:46–8.
- Li C, Sun X, Zhao K, Meng F, Li L, Mu Z, et al. Efficacy and safety of Janus kinase inhibitors for the treatment of atopic dermatitis: a systematic review and meta-analysis. Dermatology. 2022;238(4):725–35.
- Wan H, Jia H, Xia T, Zhang D. Comparative efficacy and safety of abrocitinib, baricitinib, and upadacitinib for moderate-to-severe atopic dermatitis: a network meta-analysis. Dermatol Ther. 2022;35(9): e15636.
- 57. Silverberg JI, Thyssen JP, Simpson EL, Yosipovitch G, Ständer S, Valdez H, et al. Impact of oral abrocitinib monotherapy on patient-reported symptoms and quality of life in adolescents and adults with moderate-to-severe atopic dermatitis: a pooled analysis of patient-reported outcomes. Am J Clin Dermatol. 2021;22(4):541–54.
- Cork MJ, McMichael A, Teng J, Valdez H, Rojo R, Chan G, et al. Impact of oral abrocitinib on signs, symptoms and quality of life among adolescents with moderate-to-severe atopic dermatitis: an analysis of patient-reported outcomes. J Eur Acad Dermatol Venereol. 2022;36(3):422–33.